Author’s response to reviews

Title: Community pharmacist intervention in depressed primary care patients (PRODEFAR study): randomized controlled trial protocol

Authors:

Maria Rubio-Valera (mrubio@sjd-ssm.com)
Antoni Serrano-Blanco (aserrano@sjd-ssm.com)
Pere Travé (ptrave.cp.ics@gencat.cat)
M Teresa Peñarrubia-María (32829mpm@comb.es)
Mar Ruiz (mruizfalques@gmail.com)
Marian March Pujol (marianmarch@ub.edu)

Version: 2 Date: 4 August 2009

Author’s response to reviews:

Dear BioMed Central Editorial Production Team,

The following formatting changes have been made:

Major revisions:
- References 2 and 6: Issue number has been included in both references.
- Tables 1, 3 and 4: New tables have been created using the MS Word 2007 Table Tool.

As I use the Spanish version of the software I have applied the MS Word auto-format template “básico 1” which I suppose is the equivalent to “simple 1”, if not, please let me now and I will change it.
- Table 2: Symbol crosses have been replaced with text: “X”

Minor revisions:
- “Methods” have been changed to “Methods/design” in the abstract as well as in the main text.
- Table 2 legend has been placed below the table.
- The citation that appeared as “254” in the text has been corrected.

After checking the manuscript no other changes have been made.

Please don’t hesitate to contact me if any other changes are required or if there is something wrong with the changes made.

With best wishes,

Maria Rubio Valera